Author's response to reviews

Title: p53 expression is significantly correlated with high risk of malignancy and epithelioid differentiation in gastrointestinal stromal tumors. An immunohistochemical study with 104 GISTs.

Authors:

Ursula Pauser (upausergmx.de)
Nina Schmedt auf der Günne (ninaschmedtadgyahoo.de)
Günter Klöppel (gkloeppelpathuni-kiel.de)
Hartmut Merz (merzpatho.uni-luebeck.de)
Alfred C Feller (fellerpatho.uni-luebeck.de)

Version: 5 Date: 11 July 2008

Author's response to reviews:

Dear Editorial Production Team,

enclosed you will find the revised manuscript with changes to the format which you had requested.

Sincerely yours

Dr Ursula Pauser